Author’s response to reviews

Title: Effects of patient- and system-level health literacy on patient-reported outcomes: a representative statewide survey

Authors:

Kimberly A Kaphingst (kaphingstk@wudosis.wustl.edu)
Nancy L Weaver (weavernl@mac.com)
Ricardo J Wray (wray@slu.edu)
Melissa LR Brown (mrichert@gmail.com)
Trent Buskirk (tbuskirk@slu.edu)
Matthew W Kreuter (mkreuter@brownschool.wustl.edu)

Version: 2  Date: 22 July 2014

Author’s response to reviews:

Our manuscript was not reviewed at BMC Public Health, but we are submitting to BMC Health Services Research at the suggestion of the editorial office. We agree that BMC Health Services Research would be a good fit for this manuscript.